Author's response to reviews

Title: Somatisation: illness perspectives of asylum seeker and refugee patients from the former country of Yugoslavia

Authors:

Noelle Junod Perron (noelle.junod@hcuge.ch)
Patricia Hudelson (patricia.hudelson@hcuge.ch)

Version: 5  Date: 6 February 2006

Author’s response to reviews:

Dear Editor,

We made the requested formatting changes in the submission system and the manuscript:

- In the abstract section of the manuscript, under background, Ex-Yugoslavia was replaced by the former country of Yugoslavia.

- author list: in the submission system, initials of both authors were removed: Noelle (vorname) Junod Perron (surname) and Patricia (vorname) Hudelson (surname). In the manuscript, the * put at the end of each author’s surname was removed.

- section headings: In the manuscript, ABSTRACT was replaced by Abstract. In the manuscript, the text of each section of the abstract was put on the next line. According to the formatting model, all headings of the manuscript were also put in bold characters front size 14 and subheadings in bold characters front size 12.

- Keywords: keywords at the end of the abstract section were removed.

- references: although I selected the journal's style in Endnote, the initials still appear before the lastname. I do not know how to change this and hope that it will not be a problem for you.

I hope that these changes will suit you

Best regards

Noelle Junod Perron